Author’s response to reviews

Title: Hypocholesterolemia is an independent risk factor for depression disorder and suicide attempt in Northern Mexican population

Authors:

Marcela Segoviano (marcela_segoviano@hotmail.com)
Manuel Cárdenas (long_xue_@hotmail.com)
José Salas (jsalas_pacheco@hotmail.com)
Osmel La Llave (ollave56@yahoo.es)
Marcelo Barraza (sbmaj30@hotmail.com)
Francisco Castellanos (xavier_castellanos@hotmail.com)
Fernando Vázquez (feralaniz1@hotmail.com)
Jazmin Méndez (jazminedmehe@gmail.com)
Ernesto Miranda (ernesto.miranda@mail.com)
Oscar Arias (arias@ciencias.unam.mx)
Edna Méndez (edna_madai@hotmail.com)

Version: 3 Date: 08 Dec 2017

Author’s response to reviews:

Answers to Questions and Suggested Modifications:

1. Under your Ethics approval heading in the Declarations section, please clarify whether informed consent obtained from participants was written or verbal. In addition, please list the full, unabbreviated name of the ethics review board that approved of your study.

Answers: We have now clarified that informed consent was obtained from participants in written form. We have now listed the full unabbreviated name of the ethics review board: “the Ethics Committee for Research of the General Hospital 450, Health Secretariat of Durango” (Declarations: Page 15, lines 283-285)
2. Thank you for including a Funding heading for the Declarations section. Please also state the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Answer: we have now mentioned the role of the funding body: “Materials used in the experimental stages were obtained through FOSISS CONACYT-2015 (Grant 261453). The funding body did not participate in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.” (Declarations: Page 15, lines 290-292)

3. When submitting your revised manuscript please ensure you do so as a single clean copy without any tracked changes, colored or highlighted text, as these are no longer required at this stage of the editorial process.

Answer: Our group thanks you, we have now posted a clean copy of the manuscript.